Author's response to reviews

Title: Assessment and prevention of acute health effects of weather conditions in Europe. The PHEWE project: background, objectives, design

Authors:

Paola P Michelozzi (michelozzi@asplazio.it)
Ursula U Kirchmayer (kirchmayer@asplazio.it)
Klea K Katsouyanni (kkatsoug@cc.uoa.gr)
Annibale A Biggeri (abiggeri@ds.unifi.it)
Glenn G McGregor (glenn.mcgregor@kcl.ac.uk)
Bettina B Menne (bme@ecr.euro.who.int)
Pavlos P Kassomenos (pkassom@uoi.gr)
Hugh Ross HR Anderson (r.anderson@sghms.ac.uk)
Michela M Baccini (baccini@ds.unifi.it)
Gabriele G Accetta (g.accetta@cspo.it)
Antonis A Analitis (aanalit@med.uoa.gr)
Tom T Kosatsky (tkosatsk@santepub-mtl.qc.ca)

Version: 2 Date: 4 October 2006

Author's response to reviews:

change affiliation of one of the authors (G. Accetta), submit manuscript, as table 2 was not transferred completely before